Author's response to reviews

Title: The contribution of qualitative research in designing a complex intervention for secondary prevention of coronary heart disease in two different healthcare systems [ISRCTN24081411]

Authors:

Mairead Corrigan (m.corrigan@qub.ac.uk)
Margaret E Cupples (m.cupples@qub.ac.uk)
Susan M Smith (susmith@tcd.ie)
Molly Byrne (molly.byrne@nuigalway.ie)
Claire S Leathem (c.leathem@qub.ac.uk)
Pauline Clerkin (pauline.clerkin@nuigalway.ie)
Andrew W Murphy (andrew.murphy@qub.ac.uk)

Version: 2 Date: 21 February 2006

Author's response to reviews:

Dear Editor,
I am including a figure (Figure 1) which was omitted from the Additional Files document in my original submission.
Yours sincerely
Dr Mairead Corrigan